Author’s response to reviews

Title: Assessing the Operational Effectiveness of a Maternal and Child Health (MCH) Conditional Cash Transfer Pilot Programme in Nigeria

Authors:

CHIOMA ODUENYI (chiomaoduenyi@gmail.com)

Victor Ordu (victorordu@outlook.com)

Ugo Okoli (okoliu2@gmail.com)

Version: 7 Date: 16 Jul 2019

Author’s response to reviews:

Please clarify whether written or verbal consent was obtained from a parent or legal guardian on behalf of the participants under the age of 16. If the need for parental consent was waived by an IRB or national guidelines indicated that it is not needed in the case of adolescent mothers, please clearly state this in this section.

- Participants under the age of 18 years but married, pregnant or with a baby is considered an ‘emancipated minor’ in Nigeria hence informed that she would be contacted to participate in a personal interview and written informed consent was also received.